Author’s response to reviews

Title: Functional treatment for fractures to the base of the 5th metatarsal - Influence of fracture location and fracture characteristics

Authors:

Sebastian Baumbach (sebastian.baumbach@med.uni-muenchen.de)
Wolf Prall (Christian.Prall@med.uni-muenchen.de)
Michael Kramer (Michael_s_kramer@yahoo.de)
Mareen Braunstein (Mareen.Braunstein@med.uni-muenchen.de)
Wolfgang Böcker (Wolfgang.Boecker@med.uni-muenchen.de)
Hans Polzer (Hans.Polzer@med.uni-muenchen.de)

Version: 3 Date: 25 Nov 2017

Author’s response to reviews:

Dear Mr. Mockridge, PhD,

Honored editorial board and reviewers,

thank you very much for you final approval of our manuscript. You indicated, that the figure legend to Fig. 1 did not meet your journals policy.

According to your "authors submission guidelines": "Permission should be indicated in the figure legend, and the original source included in the reference list." The legend of Fig. 1 was changed accordingly. In case you whish the whole reference spelled out in the figure legend, please let us know and we will change it accordingly.
Thank you very much for your time and giving us the opportunity to publish with your journal!

Kind regards,

On behalf of all authors